Terahertz radiation and the skin: a review

Angelina I. Nikitkina,^a Polina Y. Bikmulina^o,^{a,b} Elvira R. Gafarova,^{a,b} Nastasia V. Kosheleva,^{a,b,c} Yuri M. Efremov[®],^{a,b} Evgeny A. Bezrukov,^d Denis V. Butnaru,^d Irina N. Dolganova^o,^{a,e,f} Nikita V. Chernomyrdin^o,^{a,g} Olga P. Cherkasova⁰,^{h,i} Arsenii A. Gavdush,^g and Peter S. Timashev⁰,^{k,k,*} ^aSechenov University, Institute for Regenerative Medicine, Moscow, Russia ^bWorld-Class Research Center "Digital Biodesign and Personalized Healthcare," Moscow, Russia ^cFederal State Budgetary Scientific Institution "Institute of General Pathology and Pathophysiology," Moscow, Russia ^dSechenov University, Institute for Urology and Reproductive Health, Moscow, Russia ^eRussian Academy of Sciences, Institute of Solid State Physics, Chernogolovka, Russia ^fBauman Moscow State Technical University, Moscow, Russia ^gRussian Academy of Sciences, Prokhorov General Physics Institute, Moscow, Russia ^hRussian Academy of Sciences, Institute of Laser Physics of the Siberian Branch, Novosibirsk, Russia ¹Novosibirsk State Technical University, Novosibirsk, Russia ^JN. N. Semenov Institute of Chemical Physics, Department of Polymers and Composites, Moscow, Russia

^kLomonosov Moscow State University, Chemistry Department, Moscow, Russia

Abstract

Significance: Terahertz (THz) radiation has demonstrated a great potential in biomedical applications over the past three decades, mainly due to its non-invasive and label-free nature. Among all biological specimens, skin tissue is an optimal sample for the application of THz-based methods because it allows for overcoming some intrinsic limitations of the technique, such as a small penetration depth (0.1 to 0.3 mm for the skin, on average).

Aim: We summarize the modern research results achieved when THz technology was applied to the skin, considering applications in both imaging/detection and treatment/modulation of the skin constituents.

Approach: We perform a review of literature and analyze the recent research achievements in THz applications for skin diagnosis and investigation.

Results: The reviewed results demonstrate the possibilities of THz spectroscopy and imaging, both pulsed and continuous, for diagnosis of skin melanoma and non-melanoma cancer, dysplasia, scars, and diabetic condition, mainly based on the analysis of THz optical properties. The possibility of modulating cell activity and treatment of various diseases by THz-wave exposure is shown as well.

Conclusions: The rapid development of THz technologies and the obtained research results for skin tissue highlight the potential of THz waves as a research and therapeutic instrument. The perspectives on the use of THz radiation are related to both non-invasive diagnostics and stimulation and control of different processes in a living skin tissue for regeneration and cancer treatment.

© The Authors. Published by SPIE under a Creative Commons Attribution 4.0 Unported License. Distribution or reproduction of this work in whole or in part requires full attribution of the original publication, including its DOI. [DOI: 10.1117/1.JBO.26.4.043005]

Keywords: terahertz technology; skin; terahertz spectroscopy and imaging; regenerative medicine; cancer detection and treatment.

Paper 200356VSSR received Nov. 2, 2020; accepted for publication Jan. 19, 2021; published online Feb. 13, 2021.

^{*}Address all correspondence to Peter S. Timashev, timashev.peter@gmail.com

1 Introduction

The skin is the layer of flexible outer tissue covering the body and functions as an interface with the environment.^{1,2} The main function of the skin is the protection from external factors, for example, against various xenobiotics and pathogens. Water and temperature balances are also regulated based on the processes in the skin. The functions and dysfunctions of the skin, including many pathological conditions, have a significant impact on both the physical health and general wellness of a person.

A large number of past studies have greatly expanded our understanding of the structure and properties of the skin, and many tools for treatment and diagnostic purposes have been developed. A significant role in such studies is played by interdisciplinary, cutting-edge approaches coming from physics and biophysics. Among such approaches, the use of terahertz (THz) radiation looks particularly promising due to some recent achievements in the field. THz radiation is an electromagnetic wave with a frequency that lies in between the infrared and microwave regions—namely, in the 0.1- to 10-THz range (1 THz = 10^{12} Hz).³ Its wavelengths and photon energies range from 3 mm to 30 μ m and from 0.41 to 41 meV, respectively. Due to the high sensitivity to biomolecules and water content and the low ionization of biological samples, THz-based methods have a great potential in biomedical research and diagnostics.^{4,5} Consequently, many efforts have been devoted to the development and application of THz methods in biomedical networks and biological fields.^{5–7}

Some specifics of THz imaging and spectroscopy, as described below, make these techniques good candidates for the skin research. THz radiation is non-ionizing and is considered to be safe for humans at low powers. THz waves are strongly absorbed by water molecules, which limits their penetration into tissues by hundreds or even tens of microns. Thus, skin tissue is the ideal target for imaging using THz radiation due to its superficial location. The skin penetration is around 0.1 to 0.3 mm depending on the THz frequency.⁸ Both the content and state (free or bound) of water in tissue could be used as the markers for skin cancer detection and diagnosis of some other skin diseases. In addition to their diagnostic potential, THz technologies demonstrate perspectives for treatment using their effects on DNA demethylation and specific expression.

This review is divided into eight sections. Section 2 is devoted to the brief overview of the THz instrumentation. Section 3 provides information on the skin structure and properties. Section 4 overviews the recent THz imaging and spectroscopic techniques used for skin studies. Section 5 addresses the possible effects of THz radiation on skin cells and the extracellular matrix (ECM). Section 6 summarizes current achievements in the diagnosis and treatment of skin cancer. Section 7 covers some other recent perspective applications of THz technology in skin-related problems. Section 8 summarizes the reviewed material and addresses the limitations and perspectives of THz technology.

2 THz Instrumentation

A variety of techniques to generate and detect THz radiation have been developed in the past few decades, and these form the basis of the spectroscopic and imaging instruments.⁹ Among the existing schemes, two general types can be distinguished depending on the generated radiation—pulsed and continuous-wave (CW).

The most common CW-radiation sources are quantum cascade lasers,^{10,11} high-speed transistors, and diodes.^{12,13} Tunable CW THz waves may be obtained by backward-wave oscillators,¹⁴ parametric conversion,¹⁵ and photomixing and frequency multiplication.¹⁶ Broadband CW-radiation can be obtained as a part of the thermal source spectrum, such as that of mercury lamps and globars.¹⁷ The detection of CW-radiation is usually implemented by pyroelectric and optoacoustic (Golay cell) detectors ¹⁸ or by Li-He cooled bolometers.¹⁹ Here, we should also mention emerging THz-wave solid-state emitters and detectors, which are based on different two-dimensional materials, such as graphene and related heterostructures, as well as on novel physical principles of operation.^{20–23}

The majority of studies on biological samples have been performed using THz pulsed radiation because it yields broader information than CW-radiation does (see below). The applied



Fig. 1 A common tissue measurement scheme for imaging and spectroscopy using TPS in the reflection mode (oblique incidence). The normal incidence and transmission mode are other widely used schemes. Reproduced from Ref. 24, CC BY-NC 4.0.

techniques are THz pulsed spectroscopy (TPS) and THz pulsed imaging (TPI); the schematic of a pulsed spectrometer is shown in Fig. 1. The pulsed radiation emitters are commonly based on photoconductive antennas (PCAs),^{25–28} although other methods exist, such as optical rectification and generation in plasma.^{29–31} Modern PCA emitters produce short sub-picosecond THz pulses, featuring only a few cycles of the THz field's oscillation and a broadband spectrum. They are accompanied by femtosecond lasers, the pulsed "pump" beam of which proceeds to the PCA-emitter that generates a THz pulse due to the photoconductivity/photoswitching effect. In turn, a PCA detector is used for THz signal detection by mixing THz radiation and the "probe" beam of the femtosecond laser with an adjustable path length. The described THz-wave generation and detection principles underlie the methods of TPS and TPI. The typical forms of a THz pulse and its spectrum acquired from a biotissue sample are demonstrated in Fig. 2.



Fig. 2 TPS of biotissue samples in reflection mode: (a) a spectroscopy unit for placing the sample; (b) reference E_r without the sample and sample E_s signals of the THz pulse spectrometer in the frequency domain; and (c) the same signals in the time domain. Reproduced from Ref. 32, CC BY 4.0.

TPS might be implemented in reflection or transmission modes, and the imaging is generally achieved by raster-scanning a sample surface with a focused THz beam.^{33–36} At the same time, intensive research has been conducted on THz multipixel cameras and holographic or coded-aperture imaging principles.^{9,37–39}

The main THz spectroscopic approaches that are generally used in biological research are Fourier transform spectroscopy (FTS), photomixing spectrometry, TPS, and TPI. FTS commonly utilizes broadband CW-sources or pulsed sources and the Michelson interferometer scheme, in which the inverse Fourier transform of the recorded interferogram is used. The photomixing spectrometer also utilizes a CW-source and contains two photomixers as a transmitter and a receiver, respectively.^{40,41} This technique is inexpensive and provides high spectral density and frequency resolution, although it requires a long measurement time. TPS is currently the most versatile technique in biological applications and is associated with the development of pulsed emitters and detectors.⁵ The technique allows for registration of the time-dependent electric field of a THz pulse (not just the power). The collected data include the amplitude and phase information in the frequency domain. Such a combination of time-domain and frequency-domain information about the THz field opens wide opportunities for TPS signal processing and data analysis, as compared with common Fourier-transform spectroscopy.

Data processing is an important step of THz spectroscopy, especially when using the TPS technology. Even when only the power spectrum is measured, the complex dielectric permittivity (or the complex refractive index) of the material can be obtained using the Kramers–Kronig relations and involving some additional assumptions.^{42,43} TPS does not require the use of Kramers–Kronig relations because both the frequency-domain amplitude and phase of the THz waveform are known. The data processing steps include preprocessing of raw signals, time-domain windowing (apodization),^{44,45} denoising ^{46–48} and then deconvolution (or inverse filtering),⁴⁷ aimed at eliminating the impact of the particular TPS response function on the measured data. In TPS, reconstruction of the sample complex dielectric permittivity or complex refractive index is an ill-posed inverse spectroscopy problem related to the minimization of a discrepancy between the experimental data and the theoretical model. Various approaches for solving this inverse problem have been suggested and introduced recently for different geometries for experiments.^{49–53} Thr final processing steps often include statistical analysis and dimensionality reduction of the observed data.^{5,54}

3 Structure and Properties of the Skin

The integumentary system is a protective barrier separating the body from the environment. It is represented by the skin and its derivatives. The skin is the largest organ of mammals, it accounts for about 16% of the body weight, and its total surface area reaches 2 $m^{2.55}$ It performs many vital functions, including thermoregulatory, metabolic, receptor, endocrine, and immune ones.^{2,56}

There are three interconnected layers of tissues in the structure of the skin (Fig. 3). Its outermost layer is the epidermis, the middle one is the dermis, and the innermost one is the hypodermis.⁵⁸ The complex dynamic organization of the skin is related to the different structure and physiological characteristics of different body areas.⁵⁹ Depending on the localization, the thickness of these layers can vary.^{60,61} For example, the eyelid has the thinnest layer of the epidermis, less than 0.5 mm, while the palms and soles have the thickest layer of the epidermis, about 1.5 mm. The thickest dermis is on the back, where it is 30 to 40 times thicker than the overlying epidermis.⁶²

The epidermis is a system of continuously renewing cells, which is based on the process of specific cell differentiation called keratinization. The morphological basis of the epidermis is the multilayer flat squamous epithelium. Its structural organization changes from the innermost basal layer to the outermost stratum corneum.⁶³ Keratinocytes account for up to 95% of epidermal cells and enable the synthesis of keratin, a filamentous protein that plays a protective role.⁶⁴ The epidermis is usually divided into four layers according to the morphology and position of keratinocytes:⁶⁵



Fig. 3 Schematic representation of the human skin structure. Reproduced from Ref. 57, CC BY 4.0.

- the layer of basal cells (stratum basale);
- the layer of spinous cells (stratum spinosum);
- the layer of granule cells (stratum granulosum); and
- the layer of keratinized corneocytes (stratum corneum).

The basal layer is formed by one row of columnar keratinocytes adjacent to the basement membrane. The distinctive features of basal cells are dark-colored oval nuclei and the presence of the melanin pigment. Basal keratinocytes adhere to each other, as well as to more superficial flattened cells, through desmosomes.⁶⁶ The presence of stem cells and the mitotic activity of the basal layer provide the continuous renewal of epithelial cells and their differentiation, with gradual lifting into overlying layers, transformation into corneocytes, and desquamation from the skin surface. The basal cell migration from the basal layer to the stratum corneum in humans takes at least 14 days, and the transit through the stratum corneum to the external epidermis requires another 14 days.⁶⁷

The prickly layer (stratum spinosum) consists of several rows of large cells of a polygonal shape attached by desmosomes in the area of numerous processes ("spines") that contain bundles of tonofilaments. Prickle keratinocytes retain the ability to reproduce through mitosis. The granular layer is formed by 1 to 3 rows of flattened spindle-shaped cells with a dark nucleus. The cytoplasm of such cells also contains tonofilaments, as well as keratohyalin granules. The clear layer is present only in the thick skin; its structure contains the eleidine protein. It consists of 1 to2 rows of flattened oxyphil cells with fuzzy borders and poorly defined organelles.⁶⁸ The stratum corneum is the outermost layer of the epidermis, and it is formed by postcellular structures, corneocytes. They do not contain nuclei and organelles and are filled with keratin filaments (tonofilaments), which gives them high mechanical strength and resistance to chemicals. In the outer parts of the layer, desmosomes break apart, and corneocytes desquamate (shed) from the surface of the epidelium.⁶⁷

The above-mentioned epidermal layers are involved in the formation of the epidermal proliferative unit. It is a self-renewing unit of the epidermis that has the shape of a hexagonal cell column. Its width is equal to the width of a single corneocyte, while its height corresponds to the thickness of the epidermis, and it includes all layers of the epidermis.⁶⁹

In addition to keratinocytes, the epidermis contains populations of non-epidermal cells (melano-cytes, Langerhans cells, and Merkel cells). Melanocytes are specialized neuroglial pigment-synthesizing cells. Their body lies in the basal layer, and the long processes continue into the more superficial layers of the epidermis. Melanocytes produce melanin and transfer it to

keratinocytes. Melanin is a black-brown or yellow-red pigment that protects the nuclear apparatus of cells from damage by ultraviolet rays. Melanin is synthesized and accumulated in melanosomes, which are transported to the processes of the melanocytes. The synthesis of melanin and its transport into epithelial cells is stimulated by melanocyte-stimulating hormone and adrenocorticotropic hormone, as well as by ultraviolet light.⁷⁰

Merkel cells are neuroendocrine cells that are associated with the afferent nerve fiber and perform the receptor function. Their body lies in the basal layer, and the processes are attached to epithelial cells of the basal and prickly layers by desmosomes. In the basal part of the cell, granules that contain a mediator that is secreted into the synaptic cleft during the mechanical deformation of the processes are accumulated.⁷¹ Langerhans cells (or intraepidermal macrophages) are of bone marrow origin, and they lie in the basal or prickly layers. They capture antigens that penetrate the epidermis, process and transport them to the lymph nodes, and present them to lymphocytes, triggering an immune response.⁷²

The dermis is the connective tissue layer of the skin that is about 0.5 to 5 mm thick and located under the epidermis. The dermis serves the trophic function, gives the skin strength, and contains its derivatives. It is composed of two layers—the papillary layer and the reticular layer. The papillary layer consists of a loose fibrous connective tissue with lymph and blood capillaries, nerve fibers, and endings. It provides the connection of the dermis with the basement membrane of the epidermis with the help of reticular fibers, elastic fibers, and special anchoring fibrils. The reticular layer of the dermis is deeper, thicker, and stronger: it is formed by dense fibrous unformed connective tissue and contains a three-dimensional network of thick bundles of collagen fibers interacting with the network of elastic fibers.⁶³

The subcutaneous fatty tissue (hypodermis) is a continuation of the dermis. The structural basis of the skin's deepest layer is the white adipose tissue and layers of loose fibrous connective tissue. The adipose tissue thickness depends on the location, gender, and nature of nutrition. For example, the hypodermis is absent in the vermillion border and eyelids, while the thinnest subcutaneous tissue is present in the neck. The subcutaneous tissue plays an important role in the body, acting as a heat insulator and a storage site for nutrients, hormones, and vitamins.⁶⁰

As in any other tissue, proteins, fats, and carbohydrates form the basis of the organic composition of the skin. Proteomic skin studies have revealed that the skin contains from 155 to 174 different proteins. The main structural proteins are collagen type I, II, III, VI, XII, and XIV and other ECM proteins (elastin, lumican, mimecan, periostin, prolargin, decorin, and laminin), keratins, and cell proteins (desmoplakin, histones, actin, myosin, vimentin, and tubulin).^{60,73}

Between the skin cells, there is an intercellular "cement," which consists of polar and nonpolar lipids.^{74,75} Lipids take part in the creation of the waterproof barrier and cell adhesion, as well as in the process of the cell desquamation. The lipid profile changes toward the surface of the epidermis. Quantitatively, ceramides account for the largest proportion (up to 50%) of the skin. They are followed by cholesterol (about 25%) and free fatty acids.⁷⁵ Carbohydrates are mainly represented by mucopolysaccharides, glycogen, and glucose. Of inorganic substances, water makes up the largest proportion of the skin. The water content in the skin varies depending on a person's age.^{76,77} The content of trace elements in the skin is low, 0.5%; the most common elements are copper, zinc, arsenic, and cobalt.

The boundary location of the skin implies its exposure to dangerous external factors.⁷⁸ Wound healing is a complex sequential process including hemostasis, inflammation, proliferation, and regulation with the participation of cytokines.⁷⁹ Violation of the normal biological response to skin damage resulting from an illness, injury, or surgery, as well as prolonged adverse effects, can lead to the development of complications. Understanding the molecular, cellular, and physiological mechanisms that govern wound healing is the key to the successful treatment of skin diseases.

Most of the skin components have dimensions of less than 0.1 mm, except for basal cells, squamous epithelium cells, and multicellular structures such as sweat ducts. Thus, they are much smaller than typical THz wavelengths. (Fig. 4). However, some other types of cells, such as adipose cells and thick bundles of collagen fibers, can cause scattering of THz waves, which should be accounted for during measurements.⁴

Nikitkina et al.: Terahertz radiation and the skin: a review



Fig. 4 Comparison of typical skin components with the THz wavelength ranging from 0.03 to 3 mm. Courtesy of I. N. Dolganova.

4 THz Spectroscopy and Imaging of the Skin

THz spectroscopy and imaging of biological tissues, including the skin, are based on the interaction of THz radiation with (predominantly) tissue water, as well as with other less-polar biomolecules, separated cells, and different structural components of tissues. The characteristic sample features are observed in the maximum or minimum values of the time-domain TPS waveform or in the frequency-domain data, such as the sample refractive index and absorption coefficient in a broad frequency range. The contrast-enhancement approaches, such as the integration technique,⁸⁰ principal component analysis, linear discriminant analysis,⁸¹ or signal complexity analysis,⁸² are often applied to improve the capabilities of THz spectroscopy and imaging.

THz spectroscopy was found to differentiate skin and muscle tissue with high sensitivity.⁸² TPS systems were successfully applied for measuring the optical properties of different skin regions.⁸³ This method is sensitive enough to differ skin samples moisturized with glycerin or lanolin.⁸⁴ The THz optical properties, such as the frequency-dependent absorption coefficient and refraction index, depend on the melanin content in the skin.⁸⁵ Usually, the THz refractive index *n* of skin decreases with the frequency (see Figs. 5 and 6), though the opposite character was found for ordinary and dysplastic nevi⁸⁶ (Fig. 6).

TPS allowed for detection of differences in the stratum corneum treated with chemicals that caused changes in the content of intracellular lipids or in the conformation of proteins.⁸⁷ Free water content in the stratum corneum can be measured with THz spectroscopy.⁸⁸ Also, it can detect the pressure changes in the skin due to different water distribution profiles ⁸⁹ and the process of water desorption itself.⁸ The diffusivity of human skin can be measured and used for spatially resolved maps of water content in the skin.^{90,91} This information can be used for observing and investigating human scar healing.⁹² Reflection-mode TPI detected the water amount for a 7-h-long period after a skin burn.⁹³ This method is also advantageous in collecting the data on a drug's spreading within the skin. Additionally, the TPS and TPI methods appeared instrumental in studying dimethyl sulfoxide and glycerol diffusion into skin tissues,^{94,95} as well as related effects of tissue immersion optical clearing at THz frequencies.^{96–98}

The important feature of skin imaging is the opportunity to detect, differentiate, and identify specific skin structures, such as glands, ECM, capillaries, and others. A combination of two tissue imaging modalities, optical coherence tomography and TPI, allowed for measurement of the diameter and THz dielectric properties of sweat ducts.⁹⁹ Moreover, the properties of the sweat ducts were similar before and after the measurements. This indicates that THz imaging is safe for skin and therefore potentially can be used in the medical field. THz spectroscopy can detect changes in the spatial structure of the skin. For instance, the absorption coefficients for the 0.2- to 1.5-THz frequency band were shown to depend on varying concentrations of collagen (in the range of 2 to 3 mg/ml) and fibroblast densities (in the range of 10⁵ to 10⁶ cells/ml) in the dermal equivalents.¹⁰⁰



Fig. 5 THz optical properties of Asian, Black, and Caucasian donor tissue models with a different content of melanin grown *in vitro* for 16 days: (a) measurements on Day 2, (b) on Day 16; (c) THz optical properties of water, skin, and collagen obtained from the double Debye model. Insets: fits from the double Debye model. Reproduced from Ref. 85, © 2019 Optical Society of America.

ECM is an important part of the skin that is responsible for cell growth and differentiation, tissue mechanics, and regeneration. One of the most pronounced components of the dermis is collagen. Collagen absorbs THz radiation, thus providing data about the tissue structure and enrolled processes.¹⁰¹ Far-infrared spectroscopy was able to detect the collagen amount and its spectral shape changes after the addition of a salt solution.¹⁰² TPS allowed for detection of different salt concentrations in the collagen layers and drug absorption in the skin.^{94,102} Using TPI, the concentration and spreading area of a drug released from a collagen–chitosan scaffold were assessed,¹⁰³ and the protein–water dynamics during proteolysis of collagen-like substrates by a matrix metalloproteinase were studied.¹⁰⁴ Some other components of the skin ECM, e.g. glycoproteins, can also be detected and used for the tissue structure mapping.¹⁰⁵

5 Effects of THz Radiation on the Skin

The THz radiation impact on biological cells, tissues, and organisms have been of importance since the creation of THz sources, yet this remains insufficiently investigated. For example, the safety limits of the power density, which are set by The International Commission on Non-Ionizing Radiation Protection, are not established for frequencies above 300 GHz.¹⁰⁶ The conducted studies, as reviewed in a number of articles,^{107–109} reveal that THz radiation can have both thermal (associated with the heating of the exposed object due to absorption) and non-thermal effects on biological objects. Like other electromagnetic waves, THz beams heat biotissues, and the extent of heating depends on the applied power, while the strong absorption by water represents an important factor related to the heating. The potential tissue heating with THz waves has been shown in several model studies.^{110–112} However, such heating is associated with CW radiation sources and high-power densities, whereas for pulsed sources the average power is generally too low to cause notable changes.



Fig. 6 THz dielectric characteristics of healthy skin (in green), ordinary (in blue), and dysplastic (in red) nevi *in vivo*. Reproduced with permission from Ref. 86, © 2015 AIP Publishing.

Another class of effects is related to non-thermal mechanisms of the THz-wave—biological system interactions. Supposedly, THz radiation can induce linear or non-linear resonance effects at the molecular level, specifically in deoxyribonucleic acid (DNA), in which local disruption of hydrogen bonds can further lead to modifications in gene expression.^{4,108,113} Thus, THz radiation can serve as a convenient and effective tool for cell activity modulation. On the one hand, the majority of the used THz intensities are not harmful to cells and do not cause any decrease in their viability.¹¹⁴ Studies performed on skin cells showed no signs of apoptosis and oxidative stress.¹¹⁵ On the other hand, there are data regarding THz-induced adipogenic differentiation of melanoma skin cancer MSCs and indications that THz can influence on the protein transcription.^{116,117} More detailed data are presented in Table 1.

The ECM component of the skin might be affected indirectly by the modification of cell activity with THz radiation. For example, wound healing was stimulated by TGF- β -induced synthesis of collagen after the irradiation.¹²⁰ This finding implies a potentially beneficial application of THz radiation for skin regeneration.

6 Skin Cancer Detection and Therapy

Skin cancer is the most common malignant disease in the world, affecting men and women of any race.¹²¹ Currently, one-third of all diagnosed types of cancer is skin cancer,¹²² and the incidence of skin cancer of all types is growing [Fig. 7(a)].

Frequency	Irradiance (mW/cm ²)	Exposure time	Object	Effects	Refs.
Broad spectrum centered at 10 THz	1	2, 6, 9 h	Mouse MSCs	Exposure of cells to THz radiation for 9 h caused changes in gene expression, whereas in response to shorter duration of exposure, the changes were less pronounced. The lipid inclusions that are a characteristic sign of MSC differentiation into adipocytes were clearly visible after 9 h of exposure.	Ref. 116
1) 2.52 THz;	1.2	1) 2 h;	Mouse	It was found that genes affected by prolonged	Ref. 117
2) 10 THz		2) 2, 12 h	MOOS	differentiated cells, i.e., for adipocytes, whereas genes differentially expressed after short (2 h) THz irradiation are characteristic of pluripotent stem cells.	
1) 10 THz; 2) 2.52 THz	1.2	1) 2 h; 2) 9 h	Mouse MSCs	 The level of expression of the shock protein genes remains unchanged after h of THz irradiation. 	Ref. 114
, -		, -		2) The level of the stress-responsive CRP gene that is activated in dying cells remains low in both the control and irradiated cells suggests the absent cellular stress response.	
0.14 THz	10, 30, 50, 70, and 100	20 min	hDF	120 h after the irradiation, the proliferative activity of the irradiated cells did not differ from the non-irradiated control. The level of NO production by irradiated fibroblasts did not differ from the NO level of non-irradiated cells. The 0.14-THz radiation of 10- to 100-mW power did not affect the functional activity of human skin fibroblasts.	Ref. 115
0.15 THz	0.4	20 min	hDF	1) No effect on cell cycle;	Ref. 118
				2) no effect on heat shock response;	
				3) increase in genome damage;	
				4) no effect on clastogenic genome damage;	
				5) no effect on telomere length; the THz radiation exposure <i>in vitro</i> caused non-thermal effects on the genome.	
2.52 THz	84.8	5, 10, 20, 40, or 80 min	hDF	Cellular temperatures increased by 3°C during all THz exposures. At the used power, radiation at 2.52 THz can generate thermal effects in mammalian cells.	Ref. 112
0.14 THz	10, 30, 50, 70, and 100	20 min	hDF	After exposure to THz radiation, the proliferative activity of the irradiated cells did not differ from the control. The level of NO production by irradiated fibroblasts did not differ from the control.	Ref. 115
0.38 and 2.52 THz	0.03 to 0.9	2 and 8 h	hDF, HaCaT cells	No DNA damage was found in HaCaT and hFB cells after irradiation.	Ref. 119
0.10 to 0.15 THz	0.4	20 min	hDF	The THz irradiation resulted in the genome damage in hDFs. No changes in the expression of proteins associated with DNA damage sensing and repair were detected, indicating that THz radiation exposure may affect genome integrity through aneugenic effects.	Ref. 118

Table 1	Effects of THz radiation on cells.



Fig. 7 (a) Incidence of different types of skin cancer, reproduced from Ref. 123, CC BY 4.0. (b) Histological sections of melanoma, BCC, SCC, and some rare subtypes of skin cancer (Kaposi's sarcoma and Merkel's cell carcinoma). Reproduced from Ref. 124, CC BY-NC 3.0.

Skin cancer is generally divided into two main types: melanoma (reportedly, the most dangerous cancer of the skin) and non-melanoma skin cancer (NMSC).¹²⁵ Melanoma develops from melanocytes that produce melanin, a pigment that stains the skin, eyes, and hair. Melanoma is one of the most aggressive and resistant to treatment types of human cancers, and it accounts for 75% of all deaths from skin cancer.¹²⁶ NMSC is the fifth most common cancer, with more than 1 million diagnoses worldwide in 2018.¹²⁷ NMSC is further divided into two types: basal cell carcinoma (BCC) and squamous cell carcinoma (SCC) [Fig. 7(B)]. Although BCC is the most common form of skin cancer, it is rarely fatal but can lead to serious health problems. SCC is the second most common form of skin cancer. Together, BCC and SCC make up approximately 95% of NMSC.¹²⁸ In addition to the above-mentioned types, there are also some rare types of skin cancer, such as Merkel's carcinoma,¹²⁹ Kaposi's sarcoma,¹³⁰ and dermatofibrosarcoma protuberans (DFSP). Merkel's carcinoma is the second leading cause of death from skin cancer after melanoma, although it causes less than 1% of malignant skin tumors.¹³¹ DFSP, which is ~1.3 to 7.5 times less common than Merkel's carcinoma, rarely metastases,¹³² and the prognosis is usually much better.

The development of new methods for the early-non-invasive and intraoperative diagnosis of skin cancer is extremely important. If skin cancer is diagnosed and treated early, it is almost a hundred percent treatable. A visual examination of the skin is usually not enough to diagnose it, and the traditional detection of histopathology (biopsy) is still the gold standard for evaluating skin cancer. However, biopsy has many disadvantages: it is painful, relatively expensive, and time-consuming and usually produces scars. In many cases, several biopsies are required to



Fig. 8 Basal cell skin cancer observed by THz imaging *in vivo*: (a) photograph of a skin sample and (b) a THz image of a skin sample, in which the central dark regions with increased absorption correspond to the tumor locus. Reproduced from Ref. 136 with permission, Copyright 2003 SPIE.

confirm the diagnosis. Therefore, non-invasive and minimally invasive methods and instruments are demanded.

Currently, there are several methods for non-invasive diagnostics of epithelial tissues based on two- or three-dimensional skin imaging, including optical coherence tomography^{133,134} and confocal microscopy.^{61,135} For the first time, using a THz imaging system to visualize a malignant skin lesion was proven by Woodward et al.^{136,137} In that study, visualization of basal cell skin cancer was achieved due to the difference between the absorption coefficients of cancerous and normal tissues ¹³⁶ (Fig. 8). Recent studies have shown that THz spectroscopy and/or spectroscopic imaging methods can identify tumors in mammary glands,¹³⁸ lungs,¹³⁹ the pancreas,¹³⁹ and the brain.^{32,140}

In Ref. 86, differentiation was demonstrated between ordinary and dysplastic human skin nevi *in vivo* using TPS, a dysplastic nevus being considered a precursor of melanoma, i.e., the zeroth stage of its development (see Fig. 6). For *ex vivo* murine skin tissues, the differences of optical properties between the normal skin and melanoma were used for selecting the borders of melanoma regions.¹⁴¹

The appearance of an oncological neoplasm leads to an increase in the blood microvasculature and, consequently, to an increase in the tissue water content. In addition to the structural changes, the content of various chemical compounds, for example, tryptophan amino acids,¹⁴² is altered in the affected areas of skin tissues. This leads to a modified spectral dependence of the reflection coefficient in the THz region, suggesting the possibility of label-free skin cancer diagnostics using reflected THz radiation.

TPI was applied to analyze the dielectric properties of human skin.¹⁴³ Using this technique, a contrast was found between BCC and normal healthy skin. THz imaging was also utilized to analyze biological tissues using frequency conversion in gold nanoparticles and integrating an infrared camera and detector.¹⁴⁴ Joseph et al.¹⁴⁵ differentiated non-cancerous and cancerous tissue areas via two-frequency THz imaging at 1.39 and 1.63 THz.

Recently, a perspective application of THz radiation in skin cancer treatment was demonstrated. It was shown in Ref. 146 that expression of certain tumor suppressor genes can be regulated by non-thermal effects of intense THz radiation, which suggests that it can have an anti-cancer effect.¹⁴⁷ Intense THz pulses downregulated the expression of epidermal differentiation genes, which are overexpressed in psoriasis and skin cancer.¹⁴⁶ Non-thermal intense THz pulses with high (1.0 μ J) or low (0.1 μ J) energy applied for 10 min to human skin caused phosphorylation of H2AX, which indicates the formation of double-stranded breaks in DNA. The latter is extremely dangerous and can lead to cell death or cancer. However, the simultaneous activation of certain tumor suppressor proteins and regulatory cell cycle proteins, such as p53, p21, p16, and p27, which slow down the cell cycle and facilitate DNA repair, suggests that DNA damage caused by a THz pulse can be quickly restored.^{148,149}

Methylation of genes that control the cell cycle and apoptosis catalyzed by DNA methyltransferases (DNMT) is a well-studied epigenetic change that causes genetic mutations leading to carcinogenesis.¹⁵⁰ DNA demethylation has been shown to reduce the risk of cancer by preventing tumor suppressor hypermethylation or metastasis.¹⁵¹ Currently, there are some commercially available nucleoside inhibitors of DNMT, such as azacitidine (5aza, Vidaza[®]) and decitabine (5azadC, DacogenTM);¹⁵⁰ however, such drugs have many side effects caused by their low specificity.¹⁵¹ Therefore, there is a need to look for alternative approaches to DNA demethylation. THz radiation can be used as an epigenetic inhibitor in the treatment of cancer due to its ability to cause targeted DNA demethylation, similar to demethylation drugs, along with an easy control of radiation energy.¹⁵² Demethylation is related to the resonant absorption of high-power THz radiation at the characteristic 1.65-THz peak associated with cancer cells and attributed to methyl-DNA bonds. Cheon et al.¹⁵² used resonant THz radiation to dissociate methyl-DNA bonds and reduce the total DNA methylation. The degree of methylation inside melanoma cells decreased by about 10% to 15%, causing the formation of 5 to 8 abasic sites [apurinic/apyrimidinic (AP) sites] per 10^5 bp, which was significantly less compared with DNA damaged by infrared radiation.¹⁵² AP sites are one of the basic indicators of DNA damage, capable of generating DNA strand breaks, lethal mutations, or cell death.¹⁵³ Therefore, it is necessary to understand whether powerful THz radiation can damage other DNA structures in addition to breaking methyl-DNA bonds.

Thermal effects of THz exposure are also applicable in cancer treatment, but they are less studied. Hyperthermic therapy is heating of tumor tissue (from 40°C to 45°C) to kill tumor cells.¹⁵⁴ It is believed that the heat tolerance of normal cells is better than that of cancer cells due to a weaker blood flow and a worse cooling in tumor areas. Additionally, hyperthermia makes cancer cells more sensitive to radiation and drug therapy.

The current achievements of THz spectroscopy in the diagnosis and treatment of skin cancer are shown in Table 2 and in Fig. 9.

7 Other Application of THz in Skin Studies

THz imaging is becoming one of the powerful tools for non-invasive diagnostics, visualizing and differentiating living, damaged, and dead tissue by changes in hydration gradients. For example, THz reflection imaging was proposed as a tool to monitor deterioration in the feet of diabetic patients. A common consequence of diabetes mellitus is the so-called "diabetic foot," or "diabetic foot syndrome." This syndrome is characterized by deterioration of micro- and macrovasculatue and innervation. This leads to an altered structure and physiology of the foot skin and underlying tissues with a decrease in skin sensitivity.¹⁶⁵ Ultimately, it is often necessary to amputate a part of a leg or a limb as a whole.¹⁰² Early diagnosis is crucial for timely initiating the therapy and preventing the risks of ulceration, infection, and amputation. The existing approaches are based on the analysis of sensitivity and thermoregulation disorders and allow one to recognize the syndrome when innervation and vascularization are already violated. A new approach for early diagnosis of the diabetic foot using TPS based on the detection of a decrease in skin hydration was developed by a team of scientists from Mexico.¹⁶⁶ Typical THz images of normal and diabetic feet demonstrated a correlation with the water content (Fig. 10). The TPS system used in this work utilized a reflection geometry for the generation of the feet images. A special platform was constructed with a chair and two high-density polyethylene windows, which were transparent at THz frequencies and used for the patients' feet placement. The THz waveforms were collected across a mesh of 22×54 pixels spaced by 5 mm for each foot.

Visualization of burn wounds *in vivo* was achieved by correlating changes in the reflected TPS signal with a change in the local water concentration in soft tissues.⁹³ The formation and dissipation of edema in and around the burn injury and the formation and evolution of the coagulation zone (highly reflective center of the burn) and a border zone of stasis (a ring of low reflecting tissue) were imaged with a high contrast in a live rat. In an other study,¹⁶⁷ analyzing the changes in both the water content and the density of discrete scattering structures within the skin layers, the authors developed TPS-based approaches to non-invasively

Frequency	Description	Features	Instrumentation	Year	Refs.
1.39, 1.63 THz	The differences between normal skin and skin afflicted with NMSC were investigated.	Visualization was performed at two frequencies of 1.39 and 1.63 THz, which are within and outside the absorption band of tryptophan, respectively.	CW THz spectroscopy, transmitted signal	2010	Ref. 142
0.1 to 2.5 THz	The dielectric properties of human skin were analyzed to differentiate normal cells from abnormal cells (dysplastic and non-dysplastic skin nevi).	The results demonstrate that TPS is potentially an effective tool for non-invasive early diagnosis of dysplastic nevi and melanomas of the skin.	TPS, reflected signal	2015	Refs. 86 and 155
0.6 to 1.8 THz	Refractive index and absorption coefficient of melanoma were higher than those of normal tissue due to higher cell density and water content in tissue slices.	The melanoma was unambiguously identified in the frequency region of 0.6 to 1.8 THz.	TPS, transmitted signal	2020	Ref. 141
0.1 to 2.0 THz	Computational study on frozen normal skin and frozen melanoma images.	Ice is 100 times more permeable by THz radiation than liquid water, permitting imaging of frozen tissues to a depth of 5.0 mm and differentiating melanoma due to reflective boundaries.	Finite-difference time-domain computational modeling	2019	Ref. 156
0.2 to 1.4 THz	Visualization of dehydrated paraffin-embedded cutaneous malignant melanoma.	Methods of mathematical morphology and edge detection were applied to TPI images to distinguish normal and cancerous tissues.	TPI, reflected signal	2019	Ref. 157
0.3 to 1.0 THz	<i>In vivo</i> visualization of pigmented skin nevi and melanoma precursors inside healthy human skin tissues based on the analysis of THz dielectric constant curves.	Ability to differentiate dysplastic and non-dysplastic nevi using TPS was demonstrated.	TPS, reflected signal	2015	Ref. 155
0.4 to 1.6 THz	Melanoma samples had a higher refractive index and absorption coefficient than artificial normal skin in the studied frequency range; the contrast increased with frequency.	Static permittivity at low frequency and slow relaxation time can be reliable classifiers to differentiate melanoma from healthy skin regardless of the cell density.	TPS attenuated total reflection	2019	Ref. 158

Nikitkina et al.: Terahertz radiation and the skin: a review

Table 2 The use of THz spectroscopy in the diagnosis and treatment of skin cancer (in the past 10 years).

April 2021 • Vol. 26(4)

		~			
Frequency	Description	Features	Instrumentation	Year	Refs.
4.2 THz	A technique for imaging human skin is suggested based on a THz-to-IR converter in combination with an IR-sensitive camera. The converter includes a system of gold nanoparticles embedded into a substance transparent in THz and IR (Teflon [®]).	The proposed technique laid a foundation for practical THz skin cancer imaging.	CW THz spectroscopy, THz-to-IR-converter	2017	Ref. 144
0.1 to 1.5 THz	Optical properties and spectral features of malignant skin melanocytes were investigated by TPS.	The difference in optical properties allowed for easy discrimination of mice's malignant melanocytes from normal cells.	TPS, transmission mode	2020	Ref. 159
0.077 THz	A low-power 77-GHz CW radar design for biomedical imaging applications was proposed.	The radar can measure the dielectric properties of tissues, making it suitable for the detection of melanomas with an accuracy on the order of tens of microns.	Low-power multitone 77-GHz CW radar using a sensor based on low-cost Miniature Hybrid Microwave Integrated Circuit design	2020	Ref. 160
0.1 to 1.3 THz	Oral malignant melanoma was visualized using THz reflection imaging at room temperature and below the water freezing point (-20°C).	The images of the frozen tissue without liquid water showed better contrast due to the greater penetration of THz radiation into the sample. Significant structural differences between malignant oral melanoma cells and normal mucosal cells supposedly caused the contrast.	2-D and B-scan (reconstructed using the time-domain waveform) THz imaging	2013	Ref. 161
0.2 to 1.6 THz	A study on artificial human skin tissues with and without metastatic melanomas.	Both the refractive indices and absorption coefficients of the artificial skin with melanomas are higher than those of the normal artificial skin in the entire frequency range, due to a higher water content in malignant tissues.	TPS	2018	Ref. 162
0.584 THz	Study on thick excess cancer specimens, NMSC (BCC) and SCC.	Cross-polarized THz images exhibited lower reflectivity values in cancer as compared with normal tissue and allowed for correct identification of the tumor location.	Combination of polarized light imaging and CW-THz imaging (reflected signal)	2012	Ref. 163

Table 2 (Continued).

April 2021 • Vol. 26(4)

Nikitkina et al.: Terahertz radiation and the skin: a review

		Table 2 (Continued).			
Frequency	Description	Features	Instrumentation	Year	Refs.
1.39, 1.63 THz	Two-dimensional THz transmission images of non-melanoma skin cancers were acquired with a spatial resolution of 0.39 mm and compared with the histology data.	The difference in transmission between the normal and cancerous tissue was found to be 60% at both the used frequencies due to differences in water content.	CW-THz imaging (transmitted signal)	2011	Ref. 145
0.2 to 2.5 THz	Exposure of artificial human skin tissue to intense picosecond THz pulses affects the expression levels of numerous genes associated with non-melanoma skin cancers, primarily the genes of the epidermal differentiation complex.	The ability of intense THz pulses to cause favorable changes in the expression of multiple genes implicated in skin cancers was demonstrated.	THz pulse exposure	2013	Ref. 146
1.7 THz	The efficiency of DNA demethylation and damage were evaluated in melanoma cells based on the parameters of pulsed THz exposure.	The degree of methylation in the melanoma cell pellets decreased by approximately 10% to 15%, demonstrating the technique's potential in cancer therapy.	THz pulse exposure	2019	Ref. 152
1.65 THz	DNA demethylation in melanoma cells using high-power THz radiation.	An effective non-invasive technique for melanoma therapy with few side effects.	Resonant high-power THz radiation	2020	Ref. 164

Nikitkina et al.: Terahertz radiation and the skin: a review



Fig. 9 The possibilities of different THz techniques to detect skin cancer types: (a), (b) reflection and transmission mode TPI, respectively; (c), (d) reflection and transmission mode CW THz imaging and spectroscopy, respectively. Here, BCC stands for basal cell carcinoma and SCC stands for squamous cell carcinoma. Courtesy of I. N. Dolganova.



Fig. 10 TPI of normal and diabetic feet and estimation of water content. Reproduced from Ref. 166, CC BY 4.0.

differentiate partial-thickness (second-degree) burns by the degree of damage. The deeper ones that require surgical intervention were distinguished from those that could naturally heal.

A reflective THz imaging system was used for visualization and quantification of the burninduced model of edema in rats.¹⁶⁸ 60 × 60 mm images were obtained in the reflective mode at the center frequency of 0.525 THz with a ~125-GHz bandwidth, using a 0.5-mm step size per ~10 min. The images and resulted data were compared with a more labor-intensive technique, depth-resolved magnetic resonance imaging, and a strong positive correlation was found. In another study,⁹² a reflection-mode TPS system was tested in scar imaging with the contrast based on changes in the refractive index and absorption coefficient of the scar; hypertrophic scars had a significantly higher refractive index than that of healthy skin, whereas normal scars had a refractive index lower than that of healthy skin. The refractive index in the scar area deviated from that of the surrounding healthy tissue even six months post-injury, which correlated well with the collagen deposition during wound healing.

The efficacy of the silicone gel sheeting strategy for skin scar repair was evaluated with THz spectroscopy.¹⁶⁹ The mechanism of the silicone gel sheeting is not well established, and investigation of the skin water amount revealed the nature of the silicone sheets-induced regeneration.

Finally, a potential application of TPS is the analysis of archaeological findings and mummified tissues. Compared with traditional x-ray and computed tomography, TPS offers a lower spatial resolution but allows for better identification of bones and cartilage with a spatial resolution on the order of 1 mm (at 0.3 THz) that is limited due to wave diffraction. In addition, TPS can provide additional information on the optical density of the sample. By changing the used THz frequency, one can vary the level of details: lower frequencies (0.24 THz) revealed large vessels in the bone tissue, and higher frequencies (0.54 THz) visualized the morphology of the bone tissue itself.¹⁷⁰

8 Conclusion

In this review, we summarized the recent developments of THz technologies related to the skin analysis, diagnosis, and treatment. The brief overview of instruments and methods demonstrates the uniqueness of information about skin tissue analyzed by THz imaging and spectroscopy.

The research results obtained in the past few decades on THz-wave–biological tissue interactions highlight several directions for further studies, though a number of limitations slow down their implementation.

Fortunately, most of the limitations associated with the small penetration depth of THz radiation in biological tissues are less dramatic in the case of the skin. However, they might be important for the analysis of the deeper skin layers. Development of compact and inexpensive THz sources is another problem that limits the technique prevalence. The spatial resolution is defined by the diffraction limit and is not enough for single-cell detection. However, several promising techniques, such as THz solid-immersion microscopy,^{35,36} can overcome this limit and demonstrate high efficiency in biomedical imaging. The new effective approaches for signal enhancement, contrast and sensitivity improvement, and general signal analysis will be helpful for detection tasks. Nevertheless, the freedom from using contrast agents makes THz-based distinction between normal and pathologically altered skin most beneficial for *in vivo* applications.

The biological effects induced by THz radiation require further study. As shown above, the effects of low-power THz radiation are not harmful, but it is capable of inducing certain biological responses at the level of gene expression. This finding opens perspectives on THz radiation use in the stimulation and control of different processes in a living skin tissue related to regeneration of damages and cancer treatment. The impact of high-power THz radiation, as well as its possible application in the destruction of cancer cells, is less studied.

Thus, the recent advances in THz technologies used to study biological tissues and, particularly, skin tissues, reveal their further potential as research and therapeutic instruments.

Disclosures

The authors have no relevant financial interests in this article and no potential conflicts of interest to disclose.

Acknowledgments

This work was financed by the Ministry of Science and Higher Education of the Russian Federation within the framework of state support for the creation and development of World-Class Research Centers "Digital Biodesign and Personalized Healthcare" 075-15-2020-926.

References

- J. Fore, "A review of skin and the effects of aging on skin structure and function," Ostomy Wound Manage 52(9), 24–35 (2006).
- G. K. Menon and A. M. Kligman, "Barrier functions of human skin: a holistic view," *Skin Pharmacol. Physiol.* 22(4), 178–189 (2009).
- T. Nagatsuma, H. Nishii, and T. Ikeo, "Terahertz imaging based on optical coherence tomography [Invited]," *Photonics Res.* 2(4), B64–B69 (2014).
- K. I. Zaytsev et al., "The progress and perspectives of terahertz technology for diagnosis of neoplasms: a review," J. Opt. 22(1), 013001 (2020).
- O. Smolyanskaya et al., "Terahertz biophotonics as a tool for studies of dielectric and spectral properties of biological tissues and liquids," *Prog. Quantum Electron.* 62, 1–77 (2018).
- 6. J. Shi et al., "Automatic evaluation of traumatic brain injury based on terahertz imaging with machine learning," *Opt. Express* **26**(5), 6371–6381 (2018).
- H. Zhao et al., "High-sensitivity terahertz imaging of traumatic brain injury in a rat model," J. Biomed. Opt. 23(3), 036015 (2018).
- 8. B. You et al., "Frequency-dependent skin penetration depth of terahertz radiation determined by water sorption-desorption," *Opt. Express* **26**, 22709 (2018).
- 9. Y. H. Tao, A. J. Fitzgerald, and V. P. Wallace, "Non-contact, non-destructive testing in various industrial sectors with terahertz technology," *Sensors* **20**(3), 712 (2020).

- D. Ushakov et al., "HgCdTe-based quantum cascade lasers operating in the GaAs phonon Reststrahlen band predicted by the balance equation method," *Opt. Express* 28(17), 25371–25382 (2020).
- 11. R. A. Khabibullin et al., "The operation of THz quantum cascade laser in the region of negative differential resistance," *Opto-Electron. Rev.* 27(4), 329–333 (2019).
- A. E. Yachmenev et al., "Arsenides-and related III–V materials-based multilayered structures for terahertz applications: various designs and growth technology," *Prog. Cryst. Growth Charact. Mater.* 66(2), 100485 (2020).
- 13. A. V. Shchepetilnikov et al., "Quantitative analysis of water content and distribution in plants using terahertz imaging," *Opt. Eng.* **59**(6), 061617 (2020).
- G. A. Komandin et al., "BWO generators for terahertz dielectric measurements," *IEEE Trans. Terahertz Sci. Technol.* 3(4), 440–444 (2013).
- K. Murate and K. Kawase, "Perspective: terahertz wave parametric generator and its applications," J. Appl. Phys. 124(16), 160901 (2018).
- R. Safian, G. Ghazi, and N. Mohammadian, "Review of photomixing continuous-wave terahertz systems and current application trends in terahertz domain," *Opt. Eng.* 58(11), 1–28 (2019).
- V. Skoromets et al., "Performance comparison of time-domain terahertz, multi-terahertz, and Fourier transform infrared spectroscopies," *J. Infrared Millimeter Terahertz Waves* 39(12), 1249–1263 (2018).
- M. J. E. Golay, "Theoretical consideration in heat and infra-red detection, with particular reference to the pneumatic detector," *Rev. Sci. Instrum.* 18(5), 347–356 (1947).
- 19. M. Shcherbatenko et al., "Nonequilibrium interpretation of DC properties of NbN superconducting hot electron bolometers," *Appl. Phys. Lett.* **109**(13), 132602 (2016).
- 20. V. Ryzhii et al., "Negative and positive terahertz and infrared photoconductivity in uncooled graphene," *Opt. Mater. Express* **9**(2), 585–597 (2019).
- 21. D. A. Bandurin et al., "Resonant terahertz detection using graphene plasmons," *Nat. Commun.* 9, 5392 (2018).
- 22. D. A. Bandurin et al., "Dual origin of room temperature sub-terahertz photoresponse in graphene field effect transistors," *Appl. Phys. Lett.* **112**(14), 141101 (2018).
- V. Ryzhii et al., "Far-infrared and terahertz emitting diodes based on graphene/black-P and graphene/MoS₂ heterostructures," *Opt. Express* 28(16), 24136–24151 (2020).
- M. El-Shenawee et al., "Cancer detection in excised breast tumors using terahertz imaging and spectroscopy," *Biomed. Spectrosc. Imaging* 8(1–2), 1–9 (2019).
- A. E. Yachmenev et al., "Metallic and dielectric metasurfaces in photoconductive terahertz devices: a review," *Opt. Eng.* 59(6), 061608 (2019).
- D. V. Lavrukhin et al., "Terahertz photoconductive emitter with dielectric-embedded high-aspect-ratio plasmonic grating for operation with low-power optical pumps," *AIP Adv.* 9(1), 015112 (2019).
- 27. D. V. Lavrukhin et al., "Shaping the spectrum of terahertz photoconductive antenna by frequency-dependent impedance modulation," *Semicond. Sci. Technol.* **34**(3), 034005 (2019).
- 28. D. S. Ponomarev et al., "Enhanced terahertz emission from strain-induced InGaAs/InAlAs superlattices," *J. Appl. Phys.* **125**(15), 151605 (2019).
- 29. R. K. Singh et al., "High power terahertz radiation generation by optical rectification of a shaped pulse laser in axially magnetized plasma," *Phys. Plasmas* 24, 103103 (2017).
- A. Schneider et al., "Generation of terahertz pulses through optical rectification in organic DAST crystals: theory and experiment," J. Opt. Soc. Am. B 23, 1822 (2006).
- 31. P. Shumyatsky and R. R. Alfano, "Terahertz sources," J. Biomed. Opt. 16(3), 033001 (2011).
- 32. A. A. Gavdush et al., "Terahertz spectroscopy of gelatin-embedded human brain gliomas of different grades: a road toward intraoperative THz diagnosis," *J. Biomed. Opt.* **24**(2), 027001 (2019).
- 33. N. V. Chernomyrdin et al., "Wide-aperture aspherical lens for high-resolution terahertz imaging," *Rev. Sci. Instrum.* **88**(1), 014703 (2017).
- 34. N. V. Chernomyrdin et al., "Solid immersion terahertz imaging with sub-wavelength resolution," *Appl. Phys. Lett.* **110**(22), 221109 (2017).

- N. V. Chernomyrdin et al., "Reflection-mode continuous-wave 0.15λ-resolution terahertz solid immersion microscopy of soft biological tissues," *Appl. Phys. Lett.* 113(11), 111102 (2018).
- 36. N. V. Chernomyrdin et al., "Numerical analysis and experimental study of terahertz solid immersion microscopy," *Opt. Eng.* **59**(6), 061605 (2019).
- H. Guerboukha, K. Nallappan, and M. Skorobogatiy, "Toward real-time terahertz imaging," *Adv. Opt. Photonics* 10(4), 843–938 (2018).
- 38. H. Guerboukha, K. Nallappan, and M. Skorobogatiy, "Exploiting k-space/frequency duality toward real-time terahertz imaging," *Optica* 5(2), 109–116 (2018).
- 39. R. I. Stantchev et al., "Real-time terahertz imaging with a single-pixel detector," *Nat. Commun.* **11**, 2535 (2020).
- 40. W. Zhang et al., "Observation of terahertz absorption signatures in microliter DNA solutions," *Appl. Phys. Lett.* **102**(2), 023701 (2013).
- S. Preu et al., "Tunable, continuous-wave terahertz photomixer sources and applications," J. Appl. Phys. 109(6), 061301 (2011).
- 42. V. Lucarini et al., *Kramers-Kronig Relations in Optical Materials Research*, Springer-Verlag, Berlin/Heidelberg (2005).
- 43. P. Silfsten et al., "Kramers–Kronig analysis on the real refractive index of porous media in the terahertz spectral range," *Opt. Lett.* **36**(5), 778–780 (2011).
- B. M. Giuliano et al., "Broadband spectroscopy of astrophysical ice analogues—I. Direct measurement of the complex refractive index of CO ice using terahertz time-domain spectroscopy," *Astron. Astrophys.* 629, A112 (2019).
- 45. A. A. Gavdush et al., "Proof of concept for continuously-tunable terahertz bandpass filter based on a gradient metal-hole array," *Opt. Express* **28**(18), 26228–26238 (2020).
- 46. M. Naftaly, "Metrology issues and solutions in THz time-domain spectroscopy: noise, errors, calibration," *IEEE Sens. J.* **13**(1), 8–17 (2013).
- Y. Chen, S. Huang, and E. Pickwell-MacPherson, "Frequency-wavelet domain deconvolution for terahertz reflection imaging and spectroscopy," *Opt. Express* 18(2), 1177–1190 (2010).
- 48. X. Chen et al., "Objective and efficient terahertz signal denoising by transfer function reconstruction," *APL Photonics* 5(5), 056104 (2020).
- 49. K. I. Zaytsev et al., "Accuracy of sample material parameters reconstruction using terahertz pulsed spectroscopy," *J. Appl. Phys.* **115**(19), 193105 (2014).
- 50. I. Pupeza, R. Wilk, and M. Koch, "Highly accurate optical material parameter determination with THz time-domain spectroscopy," *Opt. Express* **15**(7), 4335–4350 (2007).
- K. I. Zaytsev et al., "Highly accurate *in vivo* terahertz spectroscopy of healthy skin: variation of refractive index and absorption coefficient along the human body," *IEEE Trans. Terahertz Sci. Technol.* 5(5), 817–827 (2015).
- 52. G. R. Musina et al., "Optimal hyperosmotic agents for tissue immersion optical clearing in terahertz biophotonics," *J. Biophotonics* **13**(12), e202000297 (2020).
- 53. Q. Sun et al., "Exploiting total internal reflection geometry for terahertz devices and enhanced sample characterization," *Adv. Opt. Mater.* **8**(3), 1900535 (2020).
- H. Liu et al., "Dimensionality reduction for identification of hepatic tumor samples based on terahertz time-domain spectroscopy," *IEEE Trans. Terahertz Sci. Technol.* 8(3), 271– 277 (2018).
- 55. K. W. Ng and W. M. Lau, "Skin deep: the basics of human skin structure and drug penetration," in *Percutaneous Penetration Enhancers: Chemical Methods in Penetration Enhancement*, N. Dragicevic and H. Maibach, Eds., pp. 3–11, Springer-Verlag, Berlin/ Heidelberg (2015).
- C. Bangert, P. M. Brunner, and G. Stingl, "Immune functions of the skin," *Clin. Dermatol.* 29(4), 360–376 (2011).
- 57. M. Rahmati et al., "Designing multigradient biomaterials for skin regeneration," *Mater. Today Adv.* **5**, 100051 (2020).
- S. Kusuma et al., "Skin anatomy and physiology," in *Plastic and Reconstructive Surgery*, M. Siemionow and M. Eisenmann-Klein, Eds., Springer Specialist Surgery Series, pp. 161– 171, Springer, London (2010).

- 59. J. Oh et al., "Biogeography and individuality shape function in the human skin metagenome," *Nature* **514**, 59–64 (2014).
- 60. R. Wong et al., "The dynamic anatomy and patterning of skin," *Exp. Dermatol.* **25**(2), 92–98 (2016).
- M. Huzaira et al., "Topographic variations in normal skin, as viewed by *in vivo* reflectance confocal microscopy," *J. Invest. Dermatol.* 116(6), 846–852 (2001).
- 62. W. D. James, T. G. Berger, and D. M. Elston, *Andrews' Diseases of the Skin: Clinical Dermatology*, Saunders Elsevier, Philadelphia (2006).
- Y. Gilaberte et al., "Anatomy and function of the skin," Chapter 1 in *Nanoscience in Dermatology*, M. R. Hamblin, P. Avci, and T. W. Prow, Eds., pp. 1–14, Academic Press, Boston (2016).
- 64. J. McKittrick et al., "The structure, functions, and mechanical properties of keratin," *JOM* **64**(4), 449–468 (2012).
- 65. K. Kabashima et al., "The immunological anatomy of the skin," *Nat. Rev. Immunol.* **19**, 19–30 (2019).
- 66. J. Kanitakis, "Anatomy, histology and immunohistochemistry of normal human skin," *Eur. J. Dermatol.* **12**(4), 390–401 (2002).
- 67. G. K. Menon, "New insights into skin structure: scratching the surface," Adv. Drug Delivery Rev. 54, S3–S17 (2002).
- 68. A. Baroni et al., "Structure and function of the epidermis related to barrier properties," *Clin. Dermatol.* **30**(3), 257–262 (2012).
- 69. I. C. Mackenzie, "Ordered structure of the epidermis," *J. Invest. Dermatol.* 65(1), 45–51 (1975).
- 70. M. Cichorek et al., "Skin melanocytes: biology and development," *Postep. Dermatol. Alergol.* **1**(1), 30–41 (2013).
- S. R. Hansson and B. J. Hoffman, "Transient expression of a functional serotonin transporter in Merkel cells during late gestation and early postnatal rat development," *Exp. Brain Res.* 130(3), 401–409 (2000).
- M. Merad, F. Ginhoux, and M. Collin, "Origin, homeostasis and function of Langerhans cells and other langerin-expressing dendritic cells," *Nat. Rev. Immunol.* 8(12), 935–947 (2008).
- 73. L. M. Mikesh et al., "Proteomic anatomy of human skin," J. Proteomics 84, 190-200 (2013).
- 74. A. Pappas, "Epidermal surface lipids," *Dermato-Endocrinology* 1, 72–76 (2009).
- 75. J. Bouwstra and P. Honeywell-Nguyen, "Skin structure and mode of action of vesicles," *Adv. Drug Delivery Rev.* 54, S41–S55 (2002).
- S. Verdier-Sévrain and F. Bonté, "Skin hydration: a review on its molecular mechanisms," J. Cosmet. Dermatol. 6(2), 75–82 (2007).
- 77. J. M. Waller and H. I. Maibach, "Age and skin structure and function, a quantitative approach (II): protein, glycosaminoglycan, water, and lipid content and structure," *Skin Res. Technol.* **12**(3), 145–154 (2006).
- 78. F. Groeber et al., "Skin tissue engineering—*in vivo* and *in vitro* applications," *Adv. Drug Delivery Rev.* **63**(4), 352–366 (2011).
- 79. S. Guo and L. A. Dipietro, "Factors affecting wound healing," *J. Dent. Res.* **89**(3), 219–229 (2010).
- J. Y. Park et al., "Terahertz imaging of metastatic lymph nodes using spectroscopic integration technique," *Biomed. Opt. Express* 8(2), 1122–1129 (2017).
- S. Nakajima et al., "Terahertz imaging diagnostics of cancer tissues with a chemometrics technique," *Appl. Phys. Lett.* **90**(4), 041102 (2007).
- R. Zhang et al., "Composite multiscale entropy analysis of reflective terahertz signals for biological tissues," *Opt. Express* 25(20), 23669–23676 (2017).
- 83. I. Echchgadda et al., "Using a portable terahertz spectrometer to measure the optical properties of *in vivo* human skin," *J. Biomed. Opt.* **18**(12), 120503 (2013).
- 84. D. I. Ramos-Soto et al., "Visualization of moisturizer effects in stratum corneum *in vitro* using THz spectroscopic imaging," *Appl. Opt.* **58**(24), 6581–6585 (2019).
- 85. X. G. Peralta et al., "Terahertz spectroscopy of human skin tissue models with different melanin content," *Biomed. Opt. Express* **10**(6), 2942–2955 (2019).

- 86. K. I. Zaytsev et al., "*In vivo* terahertz spectroscopy of pigmentary skin nevi: pilot study of non-invasive early diagnosis of dysplasia," *Appl. Phys. Lett.* **106**(5), 053702 (2015).
- K. Mizukoshi et al., "Terahertz wave techniques using a metal mesh for evaluating the components of the stratum corneum," *Skin Res. Technol.* 19(1), e383–e389 (2013).
- S. I. Alekseev, I. Szabo, and M. C. Ziskin, "Millimeter wave reflectivity used for measurement of skin hydration with different moisturizers," *Skin Res. Technol.* 14(4), 390–396 (2008).
- 89. J. Wang et al., "THz *in vivo* measurements: the effects of pressure on skin reflectivity," *Biomed. Opt. Express* **9**(12), 6467–6476 (2018).
- 90. Q. Sun et al., "In vivo estimation of water diffusivity in occluded human skin using terahertz reflection spectroscopy," J. Biophotonics 12(2), e201800145 (2019).
- 91. N. Bajwa et al., "Non-invasive terahertz imaging of tissue water content for flap viability assessment," *Biomed. Opt. Express* 8(1), 460–474 (2017).
- 92. S. Fan et al., "In vivo terahertz reflection imaging of human scars during and after the healing process," J. Biophotonics 10(9), 1143–1151 (2017).
- 93. P. Tewari et al., "*In vivo* terahertz imaging of rat skin burns," *J. Biomed. Opt.* **17**(4), 040503 (2012).
- 94. K. W. Kim et al., "Terahertz dynamic imaging of skin drug absorption," *Opt. Express* **20**(9), 9476–9484 (2012).
- 95. S. J. Oh et al., "Measurement depth enhancement in terahertz imaging of biological tissues," *Opt. Express* **21**(18), 21299–21305 (2013).
- 96. I. Costantini et al., "In-vivo and ex-vivo optical clearing methods for biological tissues: review," *Biomed. Opt. Express* **10**(10), 5251–5267 (2019).
- 97. A. N. Bashkatov et al., "Measurement of tissue optical properties in the context of tissue optical clearing," *J. Biomed. Opt.* 23(9), 091416 (2018).
- O. A. Smolyanskaya et al., "Glycerol dehydration of native and diabetic animal tissues studied by THz-TDS and NMR methods," *Biomed. Opt. Express* 9(3), 1198–1215 (2018).
- S. R. Tripathi et al., "Morphology of human sweat ducts observed by optical coherence tomography and their frequency of resonance in the terahertz frequency region," *Sci. Rep.* 5, 9071 (2015).
- R. Zhang et al., "Impact of cell density and collagen concentration on the electromagnetic properties of dermal equivalents in the terahertz band," *IEEE Trans. Terahertz Sci. Technol.* 8(4), 381–389 (2018).
- M. Mizuno, N. Yaekashiwa, and S. Watanabe, "Analysis of dermal composite conditions using collagen absorption characteristics in the THz range," *Biomed. Opt. Express* 9(5), 2277–2283 (2018).
- 102. M. Mizuno et al., "Far-infrared spectroscopy of salt penetration into a collagen fiber scaffold," *J. Biol. Phys.* **41**(3), 293–301 (2015).
- 103. A. A. Mahmoud and A. H. Salama, "Norfloxacin-loaded collagen/chitosan scaffolds for skin reconstruction: preparation, evaluation and *in-vivo* wound healing assessment," *Eur. J. Pharm. Sci.* 83, 155–165 (2016).
- 104. J. Dielmann-Gessner et al., "Enzymatic turnover of macromolecules generates long-lasting protein–water-coupled motions beyond reaction steady state," *Proc. Natl. Acad. Sci. U.S.A.* 111(50), 17857–17862 (2014).
- 105. P. Huang et al., "Study on glycoprotein terahertz time-domain spectroscopy based on composite multiscale entropy feature extraction method," *Spectrochim. Acta Part A* 229, 117948 (2020).
- 106. Scientific Committee on Emerging Newly Identified Health Risks, "Opinion on potential health effects of exposure to electromagnetic fields," *Bioelectromagnetics* 36(6), 480–484 (2015).
- V. I. Fëdorov et al., "The influence of terahertz radiation on the cell's genetic apparatus," J. Opt. Technol. 84(8), 509–514 (2017).
- 108. O. P. Cherkasova et al., "Effects of terahertz radiation on living cells: a review," *Opt. Spectrosc.* **128**(6), 855–866 (2020).

- M.-O. Mattsson, O. Zeni, and M. Simkó, "Is there a biological basis for therapeutic applications of millimetre waves and thz waves?" *J. Infrared Millimeter Terahertz Waves* 39(9), 863–878 (2018).
- M. Wang et al., "An overview of cancer treatment by terahertz radiation," in *IEEE MTT-S Int. Microwave Workshop Ser. RF and Wireless Technol. Biomed. and Healthcare Appl.*, pp. 1–3 (2013).
- T. T. L. Kristensen et al., "Modeling terahertz heating effects on water," *Opt. Express* 18(5), 4727–4739 (2010).
- G. J. Wilmink et al., "In vitro investigation of the biological effects associated with human dermal fibroblasts exposed to 2.52 THz radiation," *Lasers Surg. Med.* 43(2), 152–163 (2011).
- J. Preto, "Classical investigation of long-range coherence in biological systems," *Chaos* 26(12), 123116 (2016).
- B. S. Alexandrov et al., "Non-thermal effects of terahertz radiation on gene expression in mouse stem cells," *Biomed. Opt. Express* 2(9), 2679–2689 (2011).
- 115. O. Cherkasova et al., "Studying the effect of 0.14 THz radiation on human dermal fibroblasts," *AIP Conf. Proc.* **2098**(1), 020004 (2019).
- 116. J. Bock et al., "Mammalian stem cells reprogramming in response to terahertz radiation," *PLoS One* **5**(12), e15806 (2010).
- 117. B. S. Alexandrov et al., "Specificity and heterogeneity of terahertz radiation effect on gene expression in mouse mesenchymal stem cells," *Sci. Rep.* **3**, 1184 (2013).
- V. Franchini et al., "Study of the effects of 0.15 terahertz radiation on genome integrity of adult fibroblasts," *Environ. Mol. Mutagen.* 59(6), 476–487 (2018).
- 119. H. Hintzsche et al., "Terahertz radiation at 0.380 THz and 2.520 THz does not lead to DNA damage in skin cells *in vitro*," *Radiat. Res.* **179**(1), 38–45 (2012).
- K.-T. Kim et al., "High-power femtosecond-terahertz pulse induces a wound response in mouse skin," *Sci. Rep.* 3, 2296 (2013).
- H. Tsao, "Genetics of nonmelanoma skin cancer," *Arch. Dermatol.* 137(11), 1486–1492 (2001).
- T. Diepgen and V. Mahler, "The epidemiology of skin cancer," *Br. J. Dermatol.* 146(s61), 1–6 (2002).
- 123. K. Laikova et al., "Advances in the understanding of skin cancer: ultraviolet radiation, mutations, and antisense oligonucleotides as anticancer drugs," *Molecules* 24(8), 1516 (2019).
- H. Hong, J. Sun, and W. Cai, "Anatomical and molecular imaging of skin cancer," *Clin. Cosmet. Investig. Dermatol.* 1, 1–17 (2008).
- 125. C. S. L. Mueller and J. Reichrath, "Histology of melanoma and nonmelanoma skin cancer," in *Sunlight, Vitamin D and Skin Cancer*, J. Reichrath, Ed., pp. 215–226, Springer, New York (2008).
- 126. G. P. J. Guy et al., "Vital signs: melanoma incidence and mortality trends and projections— United States, 1982–2030," MMWR Morb. Mortality Weekly Rep. 64(21), 591–596 (2015).
- World Cancer Research Fund/American Institute for Cancer Research, "Skin cancer statistics, world cancer research fund," 2018, https://www.wcrf.org/dietandcancer/skin-cancer (accessed 16 September 2020).
- 128. D. K. J. Tillman and M. T. Carroll, "Topical imiquimod therapy for basal and squamous cell carcinomas: a clinical experience," *Cutis* **79**(3), 241–248 (2007).
- 129. P. Tai, "Merkel cell cancer: update on biology and treatment," *Curr. Opin. Oncol.* **20**(2), 196–200 (2008).
- J. L. Douglas et al., "Kaposi's sarcoma: a model of both malignancy and chronic inflammation," *Panminerva Med.* 49(3), 119–138 (2007).
- 131. T. L. Fitzgerald et al., "Dramatic increase in the incidence and mortality from Merkel cell carcinoma in the United States," *Am. Surg.* **81**(8), 802–806 (2015).
- 132. S. J. Miller et al., "Dermatofibrosarcoma protuberans," *J. Natl. Compr. Cancer Network* 10(3), 312–318 (2012).
- 133. A. G. Podoleanu et al., "Three dimensional OCT images from retina and skin," *Opt. Express* 7(9), 292–298 (2000).

- M. Mogensen et al., "OCT imaging of skin cancer and other dermatological diseases," J. Biophotonics 2, 442–451 (2009).
- 135. M. Ilie et al., "Current and future applications of confocal laser scanning microscopy imaging in skin oncology (Review)," *Oncol. Lett.* **17**, 4102–4111 (2019).
- V. P. Wallace et al., "Terahertz pulsed imaging of cancers," *Proc. SPIE* 4949, 353–359 (2003).
- R. M. Woodward et al., "Terahertz pulse imaging in reflection geometry of skin tissue using time-domain analysis techniques," *Proc. SPIE* 4625, 160–169 (2002).
- 138. A. J. Fitzgerald et al., "Classification of terahertz-pulsed imaging data from excised breast tissue," *J. Biomed. Opt.* **17**(1), 016005 (2012).
- 139. M.-A. Brun et al., "Terahertz imaging applied to cancer diagnosis," *Phys. Med. Biol.* 55(16), 4615–4623 (2010).
- S. Yamaguchi et al., "Brain tumor imaging of rat fresh tissue using terahertz spectroscopy," *Sci. Rep.* 6, 30124 (2016).
- 141. D. Li et al., "Detecting melanoma with a terahertz spectroscopy imaging technique," *Spectrochim. Acta Part A* 234, 118229 (2020).
- 142. C. S. Joseph et al., "Dual-frequency continuous-wave terahertz transmission imaging of nonmelanoma skin cancers," *Proc. SPIE* **7601**, 760104 (2010).
- 143. K. I. Zaytsev et al., "In vivo spectroscopy of healthy skin and pathology in terahertz frequency range," J. Phys. Conf. Ser. 584, 012023 (2015).
- 144. K. A. Moldosanov et al., "Terahertz imaging technique for cancer diagnostics using frequency conversion by gold nano-objects," *Ferroelectrics* **509**(1), 158–166 (2017).
- C. S. Joseph et al., "Continuous wave terahertz transmission imaging of nonmelanoma skin cancers," *Lasers Surg. Med.* 43(6), 457–462 (2011).
- 146. L. V. Titova et al., "Intense THz pulses down-regulate genes associated with skin cancer and psoriasis: a new therapeutic avenue?" *Sci. Rep.* **3**, 2363 (2013).
- L. V. Titova et al., "Intense picosecond THz pulses alter gene expression in human skin tissue in vivo," Proc. SPIE 8585, 85850Q (2013).
- 148. L. V. Titova et al., "Intense THz pulses cause H2AX phosphorylation and activate DNA damage response in human skin tissue," *Biomed. Opt. Express* 4(4), 559–568 (2013).
- 149. L. V. Titova et al., "Effect of intense THz pulses on expression of genes associated with skin cancer and inflammatory skin conditions," *Proc. SPIE* **8941**, 89411G (2014).
- 150. C. Desjobert et al., "Demethylation by low-dose 5-aza-2'-deoxycytidine impairs 3D melanoma invasion partially through miR-199a-3p expression revealing the role of this miR in melanoma," *Clin. Epigen.* **11**, 9 (2019).
- 151. H. Cheon et al., "Detection and manipulation of methylation in blood cancer DNA using terahertz radiation," *Sci. Rep.* **9**, 6413 (2019).
- 152. H. Cheon et al., "Effective demethylation of melanoma cells using terahertz radiation," *Biomed. Opt. Express* **10**(10), 4931–4941 (2019).
- 153. S. Singh, M. K. Singh, and P. Das, "Biosensing of solitary and clustered abasic site DNA damage lesions with copper nanoclusters and carbon dots," *Sens. Actuators B* **255**, 763–774 (2018).
- M. O. Mattsson and M. Simkó, "Emerging medical applications based on non-ionizing electromagnetic fields from 0 Hz to 10 THz," *Med. Devices* 12, 347–368 (2019).
- 155. K. I. Zaitsev et al., "Terahertz spectroscopy of pigmentary skin nevi *in vivo*," *Opt. Spectrosc.* **119**(3), 404–410 (2015).
- Z. Vilagosh, A. Lajevardipour, and A. W. Wood, "Computational phantom study of frozen melanoma imaging at 0.45 terahertz," *Bioelectromagnetics* 40(2), 118–127 (2019).
- J. Li, Y. Xie, and P. Sun, "Edge detection on terahertz pulse imaging of dehydrated cutaneous malignant melanoma embedded in paraffin," *Front. Optoelectron.* 12(3), 317–323 (2019).
- R. Zhang et al., "Dielectric and double debye parameters of artificial normal skin and melanoma," J. Infrared Millimeter Terahertz Waves 40(6), 657–672 (2019).
- 159. A. A. Goryachuk et al., "The optical properties and spectral features of malignant skin melanocytes in the terahertz frequency range," J. Phys. Conf. Ser. 735, 012073 (2016).

- 160. H. Arab et al., "Early-stage detection of melanoma skin cancer using contactless millimeter-wave sensors," *IEEE Sens. J.* 20(13), 7310–7317 (2020).
- 161. Y. C. Sim et al., "Temperature-dependent terahertz imaging of excised oral malignant melanoma," *IEEE J. Biomed. Health. Inf.* **17**(4), 779–784 (2013).
- 162. R. Zhang et al., "Experimental characterization of artificial human skin with melanomas for accurate modelling and detection in healthcare application," in *43rd Int. Conf. Infrared, Millimeter, and Terahertz Waves*, pp. 1–2 (2018).
- 163. C. S. Joseph et al., "Imaging of *ex vivo* nonmelanoma skin cancers in the optical and terahertz spectral regions optical and terahertz skin cancers imaging," *J. Biophotonics* 7(5), 295–303 (2014).
- J.-H. Son and H. Cheon, "Toward cancer treatment using terahertz radiation: demethylation of cancer cells," *Proc. SPIE* 11390, 1139002 (2020).
- 165. E. Brocco et al., "Diabetic foot management: multidisciplinary approach for advanced lesion rescue," *J. Cardiovasc. Surg.* **59**, 670–684 (2018).
- 166. G. G. Hernandez-Cardoso et al., "Terahertz imaging for early screening of diabetic foot syndrome: a proof of concept," *Sci. Rep.* 7, 42124 (2017).
- M. H. Arbab et al., "Terahertz spectroscopy for the assessment of burn injuries *in vivo*," *J. Biomed. Opt.* 18(7), 077004 (2013).
- 168. N. Bajwa et al., "Terahertz imaging of cutaneous edema: correlation with magnetic resonance imaging in burn wounds," *IEEE Trans. Biomed. Eng.* **64**(11), 2682–2694 (2017).
- 169. J. Wang et al., "In vivo terahertz imaging to evaluate scar treatment strategies: silicone gel sheeting," *Biomed. Opt. Express* **10**(7), 3584–3590 (2019).
- 170. L. Öhrström et al., "Technical note: terahertz imaging of ancient mummies and bone," *Am. J. Phys. Anthropol.* **142**(3), 497–500 (2010).

Angelina I. Nikitkina received her bachelor's degree in biotechnology from First Moscow State Medical University, Moscow, Russia. Currently, she is a master's student at Sechenov University, Faculty of Materials Science, Moscow, Russia.

Polina Y. Bikmulina obtained her bachelor's degree in biology from Lomonosov Moscow State University, Faculty of Biology, Russia, in 2019. Currently, she is a master's student at Lomonosov Moscow State University, Faculty of Biology. Also, she is a junior researcher at the World-Class Research Center "Digital Biodesign and Personalized Healthcare," Sechenov First Moscow State Medical University, Moscow, Russia.

Elvira R. Gafarova received her specialist degree from Bashkir State Medical University, Faculty of General Medicine, Ufa, Russia, in 2015. Currently, she is a PhD student at the Institute for Regenerative Medicine and a junior researcher at the World-Class Research Center "Digital Biodesign and Personalized Healthcare," Sechenov First Moscow State Medical University, Moscow, Russia.

Nastasia V. Kosheleva received her specialist degree in physiology in 2003 and her PhD in developmental biology, embryology in 2007 from Lomonosov Moscow State University, Faculty of Biology. Currently, she is a leading researcher at the World-Class Research Center "Digital Biodesign and Personalized Healthcare," Sechenov First Moscow State Medical University. She is also a leading researcher at the Institute of General Pathology and Pathophysiology, Laboratory of Cell Biology and Developmental Pathology.

Yuri M. Efremov received his specialist degree in biophysics in 2011 and his PhD in biophysics in 2014 from Lomonosov Moscow State University, Faculty of Biology, Moscow, Russia, and did postdoctoral training at Purdue University, School of Mechanical Engineering, West-Lafayette, Indiana, USA. Currently, he is an associate professor at the World-Class Research Center "Digital Biodesign and Personalized Healthcare," Sechenov First Moscow State Medical University, Moscow, Russia.

Evgeny A. Bezrukov received his PhD in 2002 and his MD in 2008. Currently, he is a professor and head of the Urological Department Number 1 at the Institute for Urology and Reproductive

Health (Sechenov University, Moscow, Russia). His field of scientific interest is urology and oncology, laparoscopic and robotic surgery. He is a principal investigator at Cambridge-Sechenov Prostate Research Study, focusing on prostate cancer metabolomics and new biomarkers.

Denis V. Butnaru received his PhD in 2008 (thesis: "The effects of vitamins, antioxidants, and phytoestrogens on risk of the development of prostate cancer"). From 2012 to 2015, he was the head of the Department of Reconstructive Uronephrology; from 2015 to 2016, he was the deputy director of research for the Uronephrology Institute; from 2016 to 2019, he was the director of the Institute for Regenerative Medicine; and from 2019 until now, he has been a vice-rector for research at Sechenov University.

Irina N. Dolganova is a research assistant in the Laboratory of Shaped Crystals of the Institute of Solid State Physics of RAS (Chernogolovka) and in the Institute for Regenerative Medicine of Sechenov University (Moscow). She received her PhD in 2017. Her current research interests include medical instruments and optical and THz technologies in medical diagnosis and treatment. She is a member of SPIE.

Nikita V. Chernomyrdin is a researcher at the Laboratory of Broadband Dielectric Spectroscopy at Prokhorov General Physics Institute of RAS. He is also a researcher at the Institute for Regenerative Medicine at Sechenov University. His research interests include high-spatial-resolution THz imaging, TPS, and biomedical applications of THz technology.

Olga P. Cherkasova is a professor at Novosibirsk State Technical University and the head of the Laboratory of Biophysics at the Institute of Laser Physics of the Russian Academy of Sciences, Siberian Branch. Her research interests include interaction of THz radiation with biological objects and application of THz spectroscopy for diabetes and oncology diagnosis.

Arsenii A. Gavdush is currently a PhD student at BMSTU and a junior researcher at GPI RAS. His research interests include TPS and its applications, especially in biomedicine and laboratory astrophysics, inverse ill-posed problems, and signal processing.

Peter S. Timashev graduated from Lomonosov Moscow State University of Fine Chemical Technologies (Moscow, Russia). He received his PhD from Karpov Institute of Physical Chemistry (Moscow, Russia) in 2004 and DSc degree in 2016. He is the director of both Sechenov Biomedical Science and Technology Park and the Institute for Regenerative Medicine (Sechenov University, Moscow, Russia). He is an author of more than 200 publications and 11 patents and is a laureate of the Moscow Government Prize.